Author's response to reviews

Title: Calcified, Minimally Fat-Contained Angiomyolipoma Clinically Indistinguishable from a Renal Cell Carcinoma

Authors:

Jin-Li Chen (j0921713355@yahoo.com.tw)
Sheng-Tang Wu (gu20283@yahoo.com.tw)
En Meng (gu203110@yahoo.com.tw)
Shou-Hung Tang (tansohorn@yahoo.com.tw)
Chih-Wei Tsao (weisurger@gmail.com)
Guang-Huan Sun (ghsun@mail.ndmctsgh.edu.tw)
Dah-Shyong Yu (yuds@ms21.hinet.net)
Sun-Yran Chang (dap77@tpech.gov.tw)
Tai-Lung Cha (tlcha@mail.ndmctsgh.edu.tw)

Version: 2 Date: 10 March 2013

Author's response to reviews: see over
Reply to the reviewer 1:

Thank you for the comments and we have made modifications as described below:

1. The title of the manuscript is modified to “Calcified, Minimally Fat-Contained Angiomyolipoma Clinically Indistinguishable from a Renal Cell Carcinoma” since the word “mimics” might be misleading.

2. We also understand that cases of renal angiomyolipomas of small amount of fat content have been reported. Epithelioids AMLs were also not uncommon. However, in this particular case, it is more likely for educational purpose. Besides, combination of calcifications and negativity for fat in all image studies is another rare situation.

3. Thank you for the comment regarding pathology features. We have indicated this case as having an epithelioid variant of AML (at the end of case presentation). The IHC staining results were put in the text. However, we would like to be more focused on the preoperative imaging, and shorten the pathology findings of an AML.

4. Thank you for the correction. We have rewritten this paragraph. What we tried to say is that even RCCs can be fat-contained.
Reply to the reviewer 2:

Thank you for your kind comments. We really appreciate.